Author's response to reviews

Title: SDF-1alpha concentration dependent modulation of RhoA and Rac1 modifies breast cancer and stromal cells interaction

Authors:

Jennifer Pasquier (jep2026@qatar-med.cornell.edu)
Nadine Abu-Kaoud (naa2037@qatar-med.cornell.edu)
Houari Abdesselem (hoa2004@qatar-med.cornell.edu)
Aisha Madani (aym2003@qatar-med.cornell.edu)
Fabien Vidal (dox_94@yahoo.fr@qatar-med.cornell.edu)
Hamda Al Thawadi (res-te2@qatar-med.cornell.edu)
Jessica Hoarau-Véchot (jeh2036@qatar-med.cornell.edu)
Bettina Couderc (Couderc.Bettina@claudiusregaud.fr)
Gilles Favre (favre.gilles@claudiusregaud.fr)
Arash Rafii (jat2021@qatar-med.cornell.edu)

Version: 6
Date: 10 June 2015

Author's response to reviews: see over
Dear Editor,

Please find enclose our revised manuscript entitled

“SDF-1alpha concentration dependent modulation of RhoA and Rac1 modifies breast cancer and stromal cells interaction”

We corrected the editor comments,

1. The name of the ethics committee who approved the bone marrow samples is “The Hotel Dieu IRB” and it has been clarified in the methods section.

2. The consent statement has been added in the Methods section.

3. The statement “there are no non-financial competing interests” has been added in the Competing Interest section.

Dr. Rafii Arash